Author’s response to reviews

Title: BRCA Mutations in a Cohort of Iraqi Patients Presenting to a Tertiary Referral Center

Authors:

Farra Chantal (cfarra@hotmail.com)
Dagher Christelle (cd19@aub.edu.lb)
Hamadeh Lama (lh64@aub.edu.lb)
Nagi El Saghir (ns23@aub.edu.lb)
Deborah Mukherji (dm25@aub.edu.lb)

Version: 3 Date: 19 Aug 2019

Author’s response to reviews:

19th August 2019

Reply to Editor

Re: BRCA Mutations in a Cohort of Iraqi Patients Presenting to a Tertiary Referral Center

Thank you for giving us an opportunity to submit a revised manuscript.

Reply to comments:

1. Availability of data and materials: This statement has been changed to: The datasets used and analyzed during the current study are available from the corresponding author on reasonable request (page 7 in Declarations section)

2. Authorship contribution statement – page 7 in declarations, this has been amended as requested: Authors’ contributions:

CF,DM,NES designed the study, CD and LH collected data, CF,CD,LH,NES and DM were responsible for data analysis, interpretation and manuscript writing.

All authors read and approved the final manuscript: CF, CD, LH, NES, DM.
3. Supplementary files have been renamed as requested and individually cited in the text

4. A final clean version of the manuscript has been uploaded

Kind regards,

Dr. Deborah Mukherji